Author's response to reviews

Title: Analysis on the equity of government investment to primary healthcare institutions during the 3-year health reform in China

Authors:

Xiaopeng Zhang (yuhuaduola@163.com)
Yuqi Xiong (xiongxiaoqi0525@163.com)
Jing Ye (392236734@qq.com)
Zhaohua Deng (642337036@qq.com)
Xinpeng Zhang (xpzhang602@163.com)

Version: 3 Date: 26 April 2012

Author's response to reviews: see over
Dear Editors:

Thank you very much for your advice. The last editor’s comments are that we should make a kindly statement in our manuscript whether the monitoring and evaluation data of the national essential drug system conducted by China’s Ministry of Health is openly available or whether you received permission (and by who) to use this.

We make some explanations as followed: About the monitoring and evaluation of the national essential drug system, we cooperated with China’s Ministry of Health and we are responsible for the some part of the project. The data of the total revenue, financial revenue and the number of people that they covered of the health sector in our manuscript are openly to the society. So we make changes in our manuscript.

Best Regards.

Yours Sincerely,

Xinping Zhang, Ph.D M.D. Professor in Health Policy and Management
School of Health and Medical Management, Tongji Medical College of Huazhong University of Science and Technology
No.13. Hangkong Road, Wuhan, Hubei province, 430030, China.
Tel: +86 2783692332; Fax: +86 2783692996
E-mail: xpzhang602@163.com